Author's response to reviews

Title: A post-trial survey to assess the impact of dissemination of results and unmasking on participants in a 13-year randomised controlled trial on age-related cataract

Authors:

Sally L Williams (sallylouise.williams@unipr.it)
Luigina Ferrigno (luigina.ferrigno@iss.it)
Giovanni Maraini (giovanni.maraini@unipr.it)
Francesco Rosmini (francesco.rosmini@iss.it)
Robert D Sperduto (bsperduto@nc.rr.com)

Version: 4 Date: 9 June 2011

Author's response to reviews: see over
9th June, 2011

MS: 2608607754995050
A post-trial survey to assess the impact of dissemination of results and unmasking on participants in a 13-year randomised controlled trial on age-related cataract Sally L Williams, Luigina Ferrigno, Giovanni Maraini, Francesco Rosmini and Robert D Sperduto

Dear Dr. Snowdon,

I have included the List of abbreviations between “Conclusions” and “Competing interests” and Figure 1 – Flow diagram after the “References”.
I hope I have done exactly what you requested. Should you have any queries contact me at your earliest convenience.

Kind regards

Sally Louise Williams
Clinical Trial Coordinator
Section of Ophthalmology
University of Parma
Italy

Tel.: +39-0521-033472
Fax: +39-0521-347005